Author's response to reviews

Title: Effectiveness of alcohol brief intervention delivered by community pharmacists: study protocol of a two-arm randomised controlled trial.

Authors:

Ranjita Dhital (ranjita.dhital@kcl.ac.uk)
Ian Norman (ian.j.norman@kcl.ac.uk)
Cate Whittlesea (cate.whittlesea@kcl.ac.uk)
Jim McCambridge (jim.mccambridge@lshtm.ac.uk)

Version: 2 Date: 11 February 2013

Author's response to reviews:

The editorial office required the following change be made to the manuscript:
‘Please remove all sources of funding from the Authors' Contributions section and place them in the Acknowledgements section.’

Description of this change:
The following sentence in the Author’s Contribution section has been deleted:
‘RD was awarded the Hugh Linstead Fellowship Award, Pharmacy Practice Research Trust, Royal Pharmaceutical Society of Great Britain for research costs for this study.’

Details on sources of funding now only appear in the Acknowledgements section of the paper.

Please note all other sections of the manuscript, including Figures 1, 2 and Appendix A remain unchanged.